Author’s response to reviews

Title: Kenyan health stakeholder views on consent and engagement processes for the re-use of hospital inpatient data to support learning on healthcare systems

Authors:

Daniel Mbuthia (danwaweu01@gmail.com)
Sassy Molyneux (smolyenux@kemri-wellcome.org)
Maureen Njue (mwandia@gmail.com)
Salim Mwalukore (smwalukore@kemri-wellcome.org)
Vicki Marsh (vmash@kemri-wellcome.org)

Version: 3 Date: 09 Dec 2018

Author’s response to reviews:

Dear Editors

Again thanks for your careful review of the re-submission of our paper METH-D-18-00054R2, titled

“Kenyan health stakeholder views on consent and engagement processes for the reuse of hospital inpatient data to support learning on healthcare systems”

Please find our responses to your requests below:

1. We have noted that Table 3 was cited in the main text of this manuscript, however no Table 3 was attached in the manuscript nor uploaded as a separate file, please revise this in your revision.

We had omitted to delete this sentence in the manuscript in an earlier revision. The table (3) referred to had been re-labelled as Figure 2, which had been included in the previous submission. Apologies for this confusion.

2. Unfortunately BMC does not accept boxes for publication. Please remove Box A and B from the additional files 3 and 4 and include the information in a different format.
We have renamed these boxes as figures in the submitted version of the manuscript and supplementary files.

3. Please rename the section heading ‘DISCUSSION: TAKING FORWARDS OUR FINDINGS’ to ‘Discussion’ in the main text.
   This has been done.

4. On uploading your revisions, please remove any tracked changes or highlighting and include only a single clean copy of the manuscript.
   This has also been done.